Author's response to reviews

Title: Measurement of Skeletal Related Events in SEER-Medicare: A Comparison of Claims-Based Methods in Men with Metastatic Prostate Cancer

Authors:

Abdulla M. Abdulhalim (aaly001@umaryland.edu)
Eberechukwu Onukwugha (eonukwug@rx.umaryland.edu)
Corinne Woods (cwoods@rx.umaryland.edu)
C. Daniel Mullins (dmullins@rx.umaryland.edu)
Young Kwok (ykwok@umm.edu)
Yi Qian (yiq@amgen.com)
Jorge Arellano (jorge.arellano@amgen.com)
Arun Balakumaran (arunb@amgen.com)
Arif Hussain (ahussain@som.umaryland.edu)

Version: 2
Date: 19 July 2014

Author's response to reviews:

July 19, 2014

Giulia Mangiameli
Executive Editor
BMC Medical Research Methodology

Dear Dr. Mangiameli,

We are pleased to submit the paper titled “Measurement of Skeletal Related Events in SEER-Medicare: A Comparison of Claims-Based Methods in Men with Metastatic Prostate Cancer” to be considered for publication in BMC Medical Research Methodology.

As requested, page and line numbers have been added to the main document.

Thank you for your consideration.

Abdulla M. Abdulhalim, BSPharm
Doctoral Candidate
University of Maryland School of Pharmacy
220 Arch Street, Office Level One
Baltimore, MD 21201
Phone: (410) 706-0908 | Fax: (410) 706-5394
Email: aaly001@umaryland.edu